Author's response to reviews

Title: Behavioral and biochemical effects of pharmacopuncture (ST 36 and ST 25) in obese rats

Authors:

Mariana C Pontes (riavet@yahoo.com.br)
Lilian C Heck (lilianheck@gmail.com)
Janice C Coelho (janice.coelho@ufrgs.br)

Version: 5 Date: 17 August 2015

Author's response to reviews: see over
Manuscript number: 1183779667149875

Manuscript Title: Behavioral and biochemical effects of pharmacopuncture (ST 36 and ST 25) in obese rats

Response to Reviewer’s comments (point-by-point)

Reviewer: Seungmin Lee

Minor Essential Revisions


2. In the Methods section (line 120 to 124) was made a correction of grammar and comma, and was wrote the full form of the Sigma-Aldrich